Author's response to reviews

Title: Facility assessment of primary care centers to assess preparedness to provide non-communicable disease services in Madhya Pradesh, India

Authors:

Abhijit P Pakhare (abhijit.cfm@aiimsbhopal.edu.in)
Sanjeev Kumar (sanjeev.cfm@aiimsbhopal.edu.in)
Rajnish Joshi (rajnish.genmed@aiimsbhopal.edu.in)

Version: 2 Date: 28 October 2013

Author's response to reviews: see over
To,
The Editor
BMC Health Service Research

**Sub:** MS: 1465898477110732, response to comments dated 25\textsuperscript{th} Oct 2013

Dear Sir,

We have made changes in manuscript as per your comments. Details are mentioned below.

<table>
<thead>
<tr>
<th>Sr No</th>
<th>Comment</th>
<th>Changes made</th>
<th>Page No</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>Please include email addresses of all authors.</td>
<td>Email addresses of all authors are added.</td>
<td>1</td>
</tr>
<tr>
<td>2</td>
<td>Human Ethics issues</td>
<td>This research was conducted as part of Skill Upgradation Workshop organized for medical officers working in government run Primary Health Centres and Community Health Centres. Questions were restricted only to assess availability of manpower, materials and equipments which are functions of Public Health Department. No question was directly related to personal attributes of medical officers. Therefore waiver of consent was sought and was approved by Institutional Human Ethics Committee of All India Institute of Medical Sciences Bhopal, India which also granted waiver of consent.</td>
<td>4</td>
</tr>
<tr>
<td>3</td>
<td>Competing interests</td>
<td>Competing interests declared</td>
<td>10</td>
</tr>
<tr>
<td>4</td>
<td>Authors’ contributions</td>
<td>Author’s contributions given.</td>
<td>10</td>
</tr>
</tbody>
</table>

Thanking you,

Rajnish Joshi